%
% This is the Latex style file for the VisHCI 2006 workshop.
% Author: John Roddick
% Adapted by Roland Goecke, 2006
%
\documentclass{CRPITStyle}   % Include this style file
\usepackage{epsfig}   % Packages to use if you wish
%\usepackage{lscape}   % 
%\usepackage{amsmath}
%\usepackage{amssymb}
\usepackage{subfigure}
\usepackage{harvard}  
\pagestyle{empty}
\thispagestyle{empty}

%
%---------------------------------------
% Start editing here
%---------------------------------------
%
\begin{document}

\title{VisHCI 2006 Latex Template Based on CRPIT Style Guide}
%%% Remove author names and affiliations for blind review %%%
\author{Roland Goecke$^{1,2}$
\and
Antonio Robles-Kelly$^{1,2}$ 
\and 
Terry Caelli$^{1,2}$
\and
REMOVE AUTHOR NAMES AND AFFILIATIONS FOR BLIND REVIEW!}
\affiliation{$^{1}$National ICT Australia, Canberra Research Laboratory \\
$^{2}$Research School of Information Sciences and Engineering, Australian 
National University \\
Canberra, Australia \\
Email:~{\tt \{roland.goecke, Antonio.robles-kelly, terry.caelli\}@nicta.com.au}}

\maketitle

%
%---------------------------------------
%
\toappear{You can use the \textbackslash toappear to add a footnote for grant 
and other information. We prefer the copyright notice below to be the last on 
the page.}

%Include the following information for the copyright statement:
\newcommand\conferencenameandplace{\emph{HCSNet Workshop on the Use of Vision in HCI (VisHCI 2006)}, Canberra, Australia}
\newcommand\volumenumber{56}
\newcommand\conferenceyear{2006}
\newcommand\editorname{R. Goecke, A. Robles-Kelly \& T. Caelli}
\toappearstandard 

%
%---------------------------------------
%
\begin{abstract}
This paper describes the manner in which papers should be formatted for papers adhering to the ACS series, \emph{Conference in Research and Practice in Information Technology}.  The abstract should be a maximum of 250 words and should clearly identify the content of the paper.  
\end{abstract}
\vspace{.1in}

\noindent {\em Keywords:} Maximum of three keywords.

%
%---------------------------------------
%
\section{Introduction}
Normal text is 10 point, Times Roman in two columns.  Page size is A4 with 
0.8in borders on all sides, two columns with 0.2 in between them.  There 
should not be a blank line between paragraphs.  If the style is being used 
properly, a natural looking gap will be produced as \LaTeX\ thinks fit.

Headings should use the heading styles as shown.  Numbering is automatic.

%
%---------------------------------------
%
\section{Heading Level 1}

%
%-----------------------------
%
\subsection{Heading Level 2}

%
%-------------------
%
\subsubsection{Heading Level 3}
Headings below level 3 should be avoided.

Tables and figures, such as Table \ref{exampleTable} and Figure \ref{exampleFigure}, should ideally be confined to one column but where this is not possible should be located at the top of a page. Each should be given a caption using the \textbackslash caption command, for example, 

%
% Here's an example table
%
% Use \begin{table*}[t] and \end{table*} for a table that spans both columns.
% Note, that such a table should be located at the top of the page.
%
\begin{table}[htb]
\begin{center}
\begin{tabular}{|l|l|}
\hline
\textbf{Type} & \textbf{Characterisation} \\
\hline
A & Reddish \\
\hline
B & Bloated \\
\hline
C & Swims fast \\
\hline
\end{tabular}
\caption{\protect\label{exampleTable} % Use this label to refer to the table in the text using \ref{exampleTable}
Caption}
\end{center}
\end{table}

%
% Here's an example figure
%
% Use \begin{figure*}[t] and \end{figure*} for a table that spans both
% columns. Note, that such a table should be located at the top of the page.
%
\begin{figure}[htb]
\setlength{\unitlength}{1mm}
\begin{center}
\subfigure{\epsfig{file=./exampleFigure.eps,height=60mm}}
\caption{\protect\label{exampleFigure} % Use this label to refer to the figure in the text using \ref{exampleFigure}
Caption}
\end{center}
\end{figure}

Citations should use the author date format only.  For example, as mentioned
by \cite{goecke2005}, ...

This point was also referred to in other works \cite{roblesKellyHancock2005,BenZvi82,Bentley86}, detailing the process..., etc.

A basic, but much abused, rule for citations is that they should not be 
referred to as nouns.  E.g.\ ``\cite{BenZvi82} talks about'' or ``In 
\cite{BenZvi82}, the process of ...'' are both wrong.
For further information regarding formats, please contact the series editor.

%
%---------------------------------------
%
\section{Submitting Camera Ready Copy}
Once your paper has been amended as required by the refereeing process, the 
source should be provided to the conference chair as required.  Do not send a 
copy to CRPIT - this is the responsibility of the program chair alone.  
Note that PDF files are the only acceptable format.  Do remember that CRPIT 
papers are A4, not US letter size (thus on the dvips you often have to specify 
{\tt -t a4}) -- see website.

Note that if errors in formatting are found they will need to be corrected 
before printing -- in extreme circumstances the paper will be left out of the 
volume.  The most common errors in using this word template are:
\begin{enumerate}
\item Not having the correct margins -- they are 0.8 inch all round.
\item Not using A4 paper format when creating the pdf file.
\end{enumerate}

%
%---------------------------------------
%
\section{References}
References should be in the Harvard author-date format, sorted alphabetically
based on the last name of the first author, examples of which are 
shown below (shown are a thesis \cite{BenZvi82}, a book \cite{Bentley86}, a 
book section \cite{caelli_bischof1997}, a conference paper \cite{goecke2005}, 
an electronic resource \cite{mysql2001} and a journal paper 
\cite{roblesKellyHancock2005}, in that order). The required formats can be 
obtained by including the {\it Harvard} package and using style {\it agsm}. 
The files {\tt harvard.sty} and {\tt agsm.bst} are available at the VisHCI 
and CRPIT websites.

\bibliographystyle{agsm}    % This is the bibliography style file. You must include this one.

%
% You can either use your own .bib file containing all required references...
% (Uncomment the line below and edit the file name, plus comment out
%  everything from \begin{thebibliography}{xx} to \end{thebibliography} below.)
%
%\bibliography{CRPITExample}

%
% ... or you can include all required bibliographic information in this file
% (Comment out \bibliography{CRPITExample} above.)
%
\begin{thebibliography}{xx}

\harvarditem{Ben-Zvi}{1982}{BenZvi82}
Ben-Zvi, J.  \harvardyearleft 1982\harvardyearright , {\em The time relational
model}, Ph.D., University of California, Los Angeles.

\harvarditem{Bentley}{1986}{Bentley86}
Bentley, J.  \harvardyearleft 1986\harvardyearright , {\em Programming pearls},
  Addison-Wesley.

\harvarditem{Caelli \harvardand\ Bischof}{1997}{caelli_bischof1997}
Caelli, T. \harvardand\ Bischof, W. \harvardyearleft 1997\harvardyearright. 
`The Role of Machine Learning in Building Image Interpretation Systems', {\em
Spatial Computing: Issues in Vision, Multimedia and Visualization 
Technologies}, Caelli, T., Bunke, H. \harvardand\ Lam, P., eds., World 
Scientific, Singapore, pp.~143--169.

\harvarditem{Goecke}{2005}{goecke2005}
Goecke, R. \harvardyearleft 2005\harvardyearright , `Audio-Video Automatic 
Speech Recognition: An Example of Improved Performance through Multimodal 
Sensor Input', {\em in Proc. NICTA-HCSNet Multimodal User Interaction Workshop 
(MMUI2005)}, Chen, F. \harvardand\ Epps, J., eds., Conferences in Research and 
Practice in Information Technology, Vol. 57, Sydney, Australia, Australian 
Computer Society, pp.~25-32.

\harvarditem{MySQL}{2001}{mysql2001}
MySQL \harvardyearleft 2001\harvardyearright , SQL Shareware Software, 
MySQL AB Co., http://www.mysql.com/. Accessed 29 Dec 2001.

\harvarditem{Robles-Kelly \harvardand\ Hancock}{2005}{roblesKellyHancock2005}
Robles-Kelly, A. \harvardand\ Hancock, E.R. \harvardyearleft 
2005\harvardyearright , `Graph Edit Distance from Spectral Seriation', {\em 
IEEE Transactions on Pattern Analysis and Machine Intelligence}, Vol.\ 27,
No.\ 3, pp.~365--378.

\end{thebibliography}


\end{document}

